Author’s response to reviews

Title: Meta-Analysis of the Efficacy and Safety of Combination of Tamsulosin plus Dutasteride compared with Tamsulosin Monotherapy in treating Benign Prostatic Hyperplasia

Authors:
Zhongbao Zhou (346894383@qq.com)
Yuanshan Cui (2268388175@qq.com)
Jitao Wu (1310264997@qq.com)
Rui Ding (3098516949@qq.com)
Tong Cai (1293705968@qq.com)
Zhenli Gao (3446395106@qq.com)

Version: 3 Date: 22 Feb 2019

Author’s response to reviews:
Feb 22, 2019

Editor-in-Chief BMC Urology

Dear editor:

Thank you for your significant suggestion, and we are now submitting the revised version titled “Meta-Analysis of the Efficacy and Safety of Combination of Tamsulosin plus Dutasteride compared with Tamsulosin Monotherapy in treating Benign Prostatic Hyperplasia” in response to your consideration for publication in BMC Urology.

We have revised the section of “availability of data and materials statement” in your declarations (page 18, line 9-11).

All authors disclosed no direct or indirect commercial financial incentive associated with this article. This paper has not been published elsewhere in whole or in part. All authors contributed
significantly to this work and have approved the final version of the manuscript. There is no ethical/legal conflicts involved in the article.

We thank you in advance for your consideration of this manuscript.

Yours Sincerely,

Corresponding Author

Zhenli Gao

Department of Urology, the Affiliated Yantai Yuhuangding Hospital of Qingdao University, NO. 20 East Yuhuangding Road, Yantai 264000, Shandong, China

Tel: 86-535-6695579
Fax: 86-535-6695579
E-mail: 3446395106@qq.com